Author’s response to reviews

Title: Core outcome sets for use in effectiveness trials involving people with bipolar and schizophrenia in a community based setting (PARTNERS2): study protocol for the development of two core outcome sets.

Authors:

Thomas Keeley (t.j.h.keeley@bham.ac.uk)
Humera Khan (H.S.Khan@bham.ac.uk)
Vanessa Pinfold (VanessaPinfold@mcpin.org)
Paula Williamson (P.R.Williamson@liv.ac.uk)
Jonathan Mathers (J.M.Mathers@bham.ac.uk)
Linda Davies (Linda.M.Davies@manchester.ac.uk)
Ruth Sayers (RuthSayers@mcpin.org)
Elizabeth England (E.J.England@bham.ac.uk)
Siobhan Reilly (S.Reilly@lancaster.ac.uk)
Richard Byng (Richard.Byng@plymouth.ac.uk)
Linda Gask (Linda.Gask@manchester.ac.uk)
Mike Clark (m.c.clark@lse.ac.uk)
Peter Huxley (p.huxley@bangor.ac.uk)
Peter Lewis (peter.lewis@bsmhft.nhs.uk)
Maximillian Birchwood (m.j.birchwood@warwick.ac.uk)
Melanie Calvert (m.calvert@bham.ac.uk)

Version: 3
Date: 5 January 2015

Author’s response to reviews: see over
Dear Sir/Madam,

Thank you for the favourable opinion on our paper titled: PARTNERS2: A protocol for the development of core outcome sets for use in effectiveness trials involving people with bipolar and schizophrenia in a community based setting.

The two changes requested by your Handling Editor have been made:

1) The title has been changed to read: “Core outcome sets for use in effectiveness trials involving people with bipolar and schizophrenia in a community based setting (PARTNERS2): study protocol for the development of two core outcome sets.”

2) Each author is mentioned individually in the Author’s Contributions section. As requested authors initials were used.

If you require anything else from us please contact me on the details below.

Yours Sincerely

Tom

**Thomas Keeley**
Research Fellow
Primary Care Clinical Sciences
School of Health and Population Sciences
The University of Birmingham
Edgbaston
B152TT

Tel: 01214148590
Mob: 07779587641
e-mail: T.J.H.Keeley@bham.ac.uk
web: [http://www.birmingham.ac.uk/thomas-keeley](http://www.birmingham.ac.uk/thomas-keeley)